\begin{table}%t2
%\centering
\par
\caption{Spectroscopic determinations of the solar photospheric phosphorus abundance.
\label{pilit}}
\begin{tabular}{ccl}
\hline\hline
\noalign{\smallskip}
$A$(P) & $\sigma$ & Reference\\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
5.43 &      &\citet{lambert68}\\
5.45 & 0.03 &\citet{lambert78}\\
5.45 & 0.04 &\citet{anders89}\\
5.45 & 0.06 &\citet{biemont94}\\
5.49 & 0.04 &\citet{berzinsh97}\\
5.45 & 0.04 &\citet{grevesse98}\\
5.36 & 0.04 &\citet{sunabboasp}\\
5.46 & 0.04 & this work\\
\noalign{\smallskip}
\hline
\end{tabular}
\end{table}